Author's response to reviews

Title: The role of a new CD44st in increasing the invasion capability of the human breast cancer cell line MCF-7

Authors:

Xin Jian Fang (jhufo@yahoo.com.cn)
Hua Jiang (lygfxj@126.com)
Xv Peng Zhao (gilocn@163.com)
Wei Mei jiang (fangruirui@yeah.net)

Version: 8 Date: 9 July 2011

Author's response to reviews:

Dear editor:

Thank you very much for your proposed changes to the manuscript. I give point-to-point revise below.

Major revisions (we require the author to make these changes)

1 Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is 'First name Initial Last name' (e.g. Joe F Bloggs)
Revised according to the proposal.

2 Figures: The first figure referenced in the manuscript text should be Figure 1. The second, figure 2, etc. Please reorder the figure citation so that they are cited in consecutive order. All figures must be cited; figures not cited will be excluded from the final manuscript
Revised according to the proposal.

3 Figures: Please provide figure titles. These should be concise and be the first sentence of the figure legend. The legend and title should be part of the manuscript file after the reference list
Revised according to the proposal.

4 Background: Please change the title "Introduction" to "Background"
Revised according to the proposal.

All the corrections were marked in red.

5 Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our
online figure guide contains full details for preparing files for submission and can be viewed.
Corrected accordingly.